\begin{thebibliography}{10}

\bibitem{McKee}
S.~McKee, W.~A. Wulf, and T.~C. Landon.
\newblock {Bounds on Memory Bandwidth in Streamed Computations}.
\newblock In {\em {Lecture Notes in Computer Science 966}}, pages 83--99.
  Springer-Verlag, Berlin, 1995.

\bibitem{McCalpin}
John~D. McCalpin.
\newblock {Memory Bandwidth and Machine Balance in Current High Performance
  Computers}.
\newblock {\em IEEE Computer Society Technical Committee on Computer
  Architecture (TCCA) Newsletter}, December 1995.
\newblock \\http://www.computer.org/tab/tcca/news/dec95/dec95\_mccalpin.ps.

\bibitem{LINPACK1}
J.~Dongarra.
\newblock {The LINPACK Benchmark: An Explanation}.
\newblock {\em Supercomputing}, pages 10--14, Spring 1988.

\bibitem{LINPACK2}
J.~Dongarra.
\newblock {Performance of Various Computers Using Standard Linear Equations
  Software in a Fortran Environment}.
\newblock Technical Report TR MCSRD 23, Argonne National Laboratory, March
  1988.

\bibitem{Marshall97a}
J.~Marshall, C.~Hill, L.~Perelman, and A.~Adcroft.
\newblock {Hydrostatic, quasi-hydrostatic and non-hydrostatic ocean modeling}.
\newblock {\em J. Geophys. Res.}, in press 1997.

\bibitem{Klinger96}
B.~Klinger, J.~Marshall, and U.~Send.
\newblock Representation and parameterisation of deep convective plumes by
  mixing.
\newblock {\em J. Geophys. Res.}, 101 (C8):18175--18182, August 1996.

\bibitem{Speer95}
K.~G. Speer and J.~Marshall.
\newblock The growth of convective plume at seafloor hot springs.
\newblock {\em J. Marine Res.}, 53:1025--1057, 1995.

\bibitem{Visbeck96}
M.~Visbeck, J.~Marshall, and H.~Jones.
\newblock On the dynamics of convective chimneys in the ocean.
\newblock {\em J. Phys. Oceanogr.}, 26:1721--1734, September 1996.

\bibitem{Jones96}
H.~Jones and J.~Marshall.
\newblock Restratification after deep convection.
\newblock {\em J. Phys. Oceanogr.}, in press 1997.

\bibitem{Haine96}
T.~Haine and J.~Marshall.
\newblock Gravitational, symmetric and baroclinic instability of the oceanic
  mixed layer.
\newblock {\em J. Phys. Oceanogr.}, in press 1997.

\bibitem{Menem96}
D.~Menemenlis, T.~Webb, C.~Wunsch, U.~Send, and C.~Hill.
\newblock {Basin-Scale Ocean Circulation from Combined Altimetric, Tomographic
  and Model Data}.
\newblock {\em Nature}, in press 1997.

\bibitem{men97}
D.~Menemenlis and C.~Wunsch.
\newblock {Linearization of an Oceanic Circulation Model for Data Assimilation
  and Climate Studies}.
\newblock {\em Journal of Atmospheric and Oceanic Technology}, in press 1997.

\bibitem{adc96}
A.~Adcroft, C.~Hill, and J.~Marshall.
\newblock {On the Treatment of Topography in Numerical Ocean Models}.
\newblock {\em Mon. Wea. Rev.}, in press 1997.

\bibitem{Marshall97b}
J.~Marshall, A.~Adcroft, C.~Hill, and L.~Perelman.
\newblock {A finite-volume, incompressible Navier-Stokes model for studies of
  the ocean on parallel computers}.
\newblock {\em J. Geophys. Res.}, in press 1997.

\bibitem{Harlow65}
F.~H. Harlow and J.~E. Welch.
\newblock Numerical calculation of time-dependent viscous incompressible flow.
\newblock {\em Phys. Fluids}, 8:2182, 1965.

\bibitem{Brugge91}
R.~Brugge, H.~L. Jones, and J.~Marshall.
\newblock Non-hydrostatic ocean modelling for studies of open-ocean deep
  convection.
\newblock In {\em {Proceedings of the Workshop on Deep Convection and Deep
  Water Formation in the Ocean}}, pages 325--340. Elsevier Science Publishers,
  Holland, 1991.

\bibitem{Miller74b}
M.~J. Miller.
\newblock On the use of pressure as vertical co-ordinate in modelling.
\newblock {\em Quart. J. R. Meteorol. Soc.}, 100:155--162, 1974.

\bibitem{Arvind95}
A.~Shaw, K-C Cho, C.~Hill, R.~P. Johnson, J.~Marshall, and Arvind.
\newblock {A comparison of implicitly parallel multithreaded and data parallel
  implementations of an ocean model based on the Navier-Stokes equations}.
\newblock Technical Report CSG Memo 364, MIT LCS, 1995.

\bibitem{TMC91}
Thinking Machine Corporation.
\newblock {\em The CM-5 Technical Summary}, 1991.

\bibitem{Fenwick95}
D.~M.~Fenwick et.al.
\newblock {The AlphaServer 8000 Series: High-end Server Platform Development}.
\newblock {\em Digital Technical Journal}, 7(1), 1995.

\bibitem{Edmondson95}
J.~H.~Edmondson et.al.
\newblock {Internal Organization of the Alpha 21164, a 300-MHz 64-bit
  Quad-issue CMOS RISC Microprocessor}.
\newblock {\em Digital Technical Journal}, 7(1), 1995.

\bibitem{CMF}
Thinking Machine Corporation.
\newblock {\em CM Fortran Language Reference Manual, V2.1}, January 1994.

\bibitem{DEC-HPF}
J.~Harris et.al.
\newblock {Compiling High Performance Fortran for Distributed-memory Systems}.
\newblock {\em {DEC Technical Journal}}, 7(3), 1995.
\newblock \\ftp://ftp.digital.com/pub/Digital/info/DTJ/v7n3\_03\_FORTRAN.ps.

\bibitem{BLAS-DESC}
C.~Lawson, R.~Hanson, D.~Kincaid, and Krogh F.
\newblock {Basic Linear Algebra Subprograms for Fortran Usage}.
\newblock {\em ACM Transactions on Mathematical Software}, 5(3):308--323, 1979.

\bibitem{DEC-DXML}
D.~P.~Manley C.~Kamath, R.~Ho.
\newblock {DXML: A High-performance Scientific Subroutine Library}.
\newblock {\em Digital Technical Journal}, 6(3), 1994.
\newblock ftp://ftp.digital.com/pub/Digital/info/DTJ/v3n6-dxml.txt.

\bibitem{PCFD95}
C.~Hill and J.~Marshall.
\newblock {Application of a parallel Navier-Stokes model to ocean circulation}.
\newblock In {\em {Parallel Computational Fluid Dynamics: Implementations and
  Results Using Parallel Computers}}, pages 545--552. Elsevier Science
  Publishers, Holland, 1995.

\bibitem{Giering}
R.~Giering and T.~Kaminski.
\newblock {Recipes for Adjoint Code Construction}.
\newblock Technical Report 212, Max-Planck-Institute for Meteorology, 1996.

\end{thebibliography}

\section{The MIT ocean-atmosphere model}
\label{s:model}

The MIT ocean-atmosphere model is built around a versatile incompressible
Navier-Stokes solver developed at MIT for the numerical study and analysis
of rotating fluids.  A hydro-dynamical kernel, designed to be sensitive to
the transition from non-hydrostatic to hydrostatic dynamics, lies at the
heart of the model, providing computational efficiency across a broad
spectrum of simulation regimes.  A review of the model's utility for
research in a range of problems related to ocean and climate dynamics can
be found in Marshall et.al.\cite{Marshall97a}.  Implementations of the
model targeted to the MPP CM-5 and to CRAY vector supercomputers are being
actively used by researchers at
MIT\cite{Klinger96,Speer95,Visbeck96,Jones96,Haine96,Menem96,men97}.

The discrete algorithm uses a finite volume\cite{adc96}, 
time-stepping formulation
which is rich in data parallelism and can handle the intricate, order one
variations in geometry of ocean basins.  The CM-5 code achieves sustained
floating point throughput of over 10 GFlop/s on a 512 node CM-5.  Daily
production runs of the model sustain between 500 MFlop/s and 1 GFlop/s on a
32 node CM-5, and between 100 MFlop/s and 200 MFlop/s on a CRAY Y-MP.

\subsection{The numerical scheme}

The numerical procedure used in the model is described in detail in
Marshall et.al.\cite{Marshall97b} The scheme is a variant on the theme set
out by Harlow and Welch\cite{Harlow65}, in which a pressure correction to
the velocity field is used to guarantee non-divergence.

The numerical kernel is a discrete form of the Boussinesq incompressible Navier
Stokes fluid equations: it can be written in semi-discrete form to second 
order in time $\Delta t$ thus:\\

\begin{equation}\label{eq-Vt}
\frac{\underline{v}^{n+1} - \underline{v}^{n}}{\Delta t}=
\underline{G}_{v}^{n+\frac{1}{2}} - \nabla p^{n+\frac{1}{2}} 
\end{equation}

\begin{equation}\label{eq-Continuity}
\nabla .\underline{v}^{n+1} = 0
\end{equation}

Equations \ref{eq-Vt} and \ref{eq-Continuity} describe how the fluid evolves
($\underline{v}=(\underline{v}_h,w)$ is the velocity in the horizontal and
the vertical) in response to forcing due to
$\underline{G}$ (representing inertial, Coriolis, metric, gravitational,
and forcing/dissipation terms) and to the pressure gradient $\nabla p$.
The same kernel algorithm is used by us to study both the 
atmosphere and the ocean - the anelastic approximation as described in
Brugge et.al.\cite{Brugge91} and in Miller\cite{Miller74b} is used
to render the same model suitable for the study of both fluids.

The pressure field required in Equations \ref{eq-Vt} and
\ref{eq-Continuity} is found by separating $p$ into hydrostatic, surface and
non-hydrostatic parts.  Vertical variations in $p$ can be computed
hydrostatically yielding $p_{hy}$ using only information in each column of
the fluid.  Horizontal variations in $p_{s}$ are diagnosed by solving an
elliptic equation which ensures non-divergent depth integrated flow:

\begin{equation}\label{eq-Ps}
\nabla_{h}. H\nabla_{h} {p_{s}} =
\nabla_{h}.\overline{\underline{G}_{v_{h}}^{n+\frac{1}{2}}}^H -
                        \nabla_{h}.\overline{\nabla_{h} {p_{hy}}^\prime}^H
\end{equation}

\noindent
( $\stackrel{\overline{\;\;\;\;\;\;}}{\;\;\;\;\;\;}\stackrel{H}{\;}$
indicates the vertical integral over the local depth, $H$, of the domain
and the subscript $_{h}$ denotes horizontal). In the non-hydrostatic limit
a three-dimensional elliptic equation is inverted for the non-hydrostatic
pressure. 
In this study we consider the hydrostatic case involving the
two-dimensional problem, Equation \ref{eq-Ps}.
More details of the numerical procedure can be found in Marshall
et.al\cite{Marshall97a,Marshall97b}.

\subsection{Program structure}

Figure {\ref{fig:FLOWDIAG}{\it } shows the high-level program structure of
the MIT model. Almost all of the execution time is spent in the
time-stepping loop, which is divided into two parts -- {\bf Forward\_Step}
and {\bf Inversion\_Step}.  The {\bf Forward\_Step} comprises a series of
nearest-neighbor computations involving two time levels of the model state
variables and their time derivatives. This stage computes explicit time
derivative terms for the velocity and tracer equations, steps them forward
in time and updates the density field through an equation of state.  There
is considerable latitude for different orderings of the computations within
{\bf Forward\_Step}, and, as discussed in Shaw et.al.\cite{Arvind95}, this
stage contains an abundance of both data parallelism and functional
parallelism.  The {\bf Inversion\_Step} requires global communication to
solve a set of simultaneous equations to ensure that the flow remains 
non-divergent.

\begin{fig}{fig:FLOWDIAG}{High-level structure of the MIT model. The 
model iterates repeatedly over a time-stepping loop, comprised of two main
blocks, {\bf Forward\_Step} and {\bf Inversion\_Step}.}
\begin{center}
\fbox{
\begin{minipage}{4in}
\sffamily
\begin{tabbing}
\hspace{.15cm} \= \hspace{.15cm} \= \hspace{.15cm} \= \hspace{3cm} \= \kill
 BEGIN \\
 \> Initialize \\
 \> \> Define topography, initialize flow field, tracer distributions\\
 \> FOR each time step DO \\
 \> \> {\sffamily\bfseries Forward\_Step} \\
 \> \> \> Calculate velocity time derivatives  \\
 \> \> \> Calculate tracer field time derivatives \\
 \> \> \> Step forward state variables \\
 \> \> \> Update density field \\
 \> \> {\sffamily\bfseries Inversion\_Step} \\
 \> \> \> Solve elliptic problem to ensure non-divergent flow\\
 \> END FOR \\
 END
\end{tabbing}
\end{minipage}
}
\end{center}
\end{fig}
